Author’s response to reviews

Title: Expression of Tissue factor in Adenocarcinoma and Squamous Cell Carcinoma of the Uterine Cervix: Implications for immunotherapy with hI-con1, a factor VII-IgG Fab chimeric protein targeting tissue factor

Authors:

Emiliano Cocco (emiliano.cocco@yale.edu)
Joyce Varughese (joyce.varughese@yale.edu)
Natalia Buza (natalia.buza@yale.edu)
Stefania Bellone (stefania.bellone@yale.edu)
Michelle Glasgow (michelle.glasgow@yale.edu)
Marta Bellone (marta.bellone@yale.edu)
Paola Todeschini (paola.todeschini@yale.edu)
Luisa Carrara (luisa.carrara@gmail.com)
Dan-Arin Silasi (dan-arin.silasi@yale.edu)
Masoud Azodi (masoud.azodi@yale.edu)
Peter E Schwartz (peter.schwartz@yale.edu)
Thomas J Rutherford (thomas.rutherford@yale.edu)
Sergio Pecorelli (s.pecorelli@aifa.gov.it)
Charles J Lockwood (charles.lockwood@yale.edu)
Alessandro D Santin (alessandro.santin@yale.edu)

Version: 5 Date: 10 June 2011

Author’s response to reviews:

To whom it may concern:

Affiliations: Each affiliation should be written in full, in the following format: Department, Institute, (Street), City, (Postal code), Country. This may mean repeating institutional addresses if the authors are from different departments within an institution. Done.

Tables: we are unable to display shading and colours within the tables. Can we therefore ask you to use another method of highlighting the shaded text e.g. by making it bold. Table have been revised.

Minor revisions (we can make these changes for you, although it will speed up publication of your manuscript if you do them while making the major changes above)

Acknowledgements: please place this section after 'Author's Contributions'.
Done.

Tables: please note that we are unable to correctly display merged cells where the merged cell crosses rows: please re-layout your table without these merged elements. Please remove the visible vertical lines from your tables. Done.
Regards,
Alessandro Santin MD